

BOSTON BOROUGH COUNCIL

Municipal Buildings, West Street, Boston, Lincolnshire, PE21 8QR

Town and Country Planning Act 1990

APPLICATION DECISION NOTICE

Application Reference: B/21/0019

Applicant: Mr G Dunmore

Rectory Farm Great Fen Road Wyberton Fen Boston

Boston PE21 7PB Agent: Mrs Jenny McIntee

JMAD Architecture 119, Northampton Road

Wellingborough

NN8 3PL

In pursuance of the powers exercised by it as Planning Authority, Boston Borough Council, having considered your application to carry out the following development:

Conversion and alteration of 2 adjoining barns into 1 single residential dwelling at Rectory Farm, Great Fen Road, Wyberton Fen, Boston, PE21 7PB

And in accordance with this notice and the particulars given in the application, do hereby give notice of its decision to **GRANT Full Planning Permission** for the said development subject to **compliance with the following condition(s):**

1. The development hereby permitted shall be begun before the expiration of four years from the date of this permission.

Reason: Required to be imposed pursuant to Section 51 of the Planning and Compulsory Purchase Act 2004.

- 2. The development hereby permitted shall be carried out in strict accordance with the application received on 03/03/2021 and in accordance with the associated plans referenced:
 - 20115 01B Site Location Plan;
 - 20115 02D Proposed Site Block Plan;
 - 20115 04E Proposed Floor Plans & Elevations;
 - 20115 05B Proposed Sections;
 - 20115 06A Landscaping and External Works Site Plan.

Reason: To ensure the development is undertaken in accordance with the approved details, in accordance with Policies 2, 3 and 4 of the South East Lincolnshire Local Plan 2011-2036 and with the intentions of the National Planning Policy Framework (2019).

3. The development permitted by this planning permission shall be carried out in accordance with the approved Flood Risk Assessment (FRA) January 2021 Version 1, RM Associates and the following mitigation measures detailed within the FRA:

.

- The ground floor living accommodation for the proposed dwelling is to be raised 500mm above the existing average ground level, set at 2.80mODN;
- Flood resilience and resistance measures to be incorporated into the proposed development as stated.

The mitigation measures shall be fully implemented prior to occupation and subsequently remain in place thereafter.

Reason: To reduce the risk of flooding to the proposed development and future occupants, in addition to providing the appropriate level of protection in accordance with the National Planning Policy Framework and Policies 2 and 4 of the South East Lincolnshire Plan 2011-2036.

4. The water consumption of each dwelling hereby permitted should not exceed the requirement of 110 litres per person per day (as set out as the optional requirement in Part G of the Building Regulations 2010 and Policy 31 of the South East Lincolnshire Local Plan 2011-2036.

The person carrying out the work must inform the Building Control Body that this duty applies.

A notice confirming the requirement for the water consumption has been met shall be submitted to the Building Control Body and Local Planning Authority, no later than five days after the completion of each individual dwelling.

Reason: To protect the quality and quantity of water resources available to the district. This condition is imposed in accordance with Policy 31 of the South East Lincolnshire Local Plan 2011-2036.

- 5. No development shall take place above ground level until details regarding the inclusion of the enhancement measures outlined in the approved Ecology and Protected Species Survey (Helen Scarborough, February 2021) have been submitted to and approved by the Local Planning Authority. The measures shall include:
 - Specification, location and number of sparrow terrace boxes and location of bird next boxes;
 Timetable for implementation of the above measures.
 - The approved works shall be carried out in accordance with the approved details.

Reason: In the interests of the natural environment in accordance with Policy 28 of the South East Lincolnshire Local Plan 2011-2036.

6. The precautionary measures outlined in the approved Ecology and Protected Species Survey (Helen Scarborough, February 2021) in regards to bats and birds will be followed at all times during the construction of the approved development.

Reason: In the interests of the natural environment in accordance with Policy 28 of the South East Lincolnshire Local Plan 2011-2036.

7. If development has not commenced within 12 months of the date of this permission, prior to the commencement of development the building(s) shall be re-surveyed for the presence of protected species and if present, a scheme of mitigation shall be submitted to the Local Planning Authority for approval in writing. The development shall be undertaken in accordance with the scheme of mitigation.

Reason: In the interest of protected species and to accord with the Wildlife and Countryside Act 1981 (as amended), the National Planning Policy Framework 2019 and Policy 28 of the South East Lincolnshire Local Plan 2011-2036.

- 8. The development hereby permitted shall not be commenced until details of a comprehensive contaminated land investigation has been submitted to and approved in writing by the Local Planning Authority and until the scope of works approved therein have been implemented. The assessment shall include all of the following measures unless the LPA dispenses with any such requirements in writing:
 - a) A Phase I desk study carried out to identify and evaluate all potential sources of contamination and the impacts on land and/or controlled waters, relevant to the site. The desk study shall establish a 'conceptual model' of the site and identify all plausible pollutant linkages. Furthermore, the assessment shall set objectives for intrusive site investigation works/ Quantitative Risk Assessment (or state if none required). A full desk top study and a non-technical summary shall be submitted in writing to the Local Planning Authority.

A site investigation shall be carried out to fully and effectively characterise the nature and extent of any land contamination and/or pollution of controlled waters. It shall specifically include a risk assessment that adopts the Source-Pathway-Receptor principle and take into account the site's existing status and proposed new use. A copy of the site investigation and findings shall be submitted in writing to the Local Planning Authority.

Reason: To ensure potential risks arising from previous site uses have been fully assessed and in accordance with Policy 30 of the South East Lincolnshire Local Plan 2011-2036.

9. Where the risk assessment (see preceding condition) identifies any unacceptable risk or risks, a detailed remediation strategy to deal with land contamination and/or pollution of controlled waters affecting the site shall be submitted to and approved in writing by the Local Planning Authority. No works, other than investigative works, shall be carried out on the site prior to receipt of written approval of the remediation strategy by the Local Planning Authority.

Reason: To ensure the proposed remediation plan is appropriate and in accordance with Policy 30 of the South East Lincolnshire Local Plan 2011-2036.

10. Remediation of the site shall be carried out in accordance with the approved remediation strategy (see preceding condition). No deviation shall be made from this scheme.

Reason: To ensure site remediation is carried out to the agreed protocol in accordance with Policy 30 of the South East Lincolnshire Local Plan 2011-2036.

11. On completion of remediation, a copy of a closure report shall be submitted to the Local Planning Authority for approval in writing. The report shall provide validation and certification that the required works regarding contamination have been carried out in accordance with the approved Method Statement(s). Post remediation sampling and monitoring results shall be included in the closure report.

Reason: To provide verification that the required remediation has been carried out to the required standards in accordance with Policy 30 of the South East Lincolnshire Local Plan 2011-2036.

12. If, during development, contamination not previously considered is identified, then the Local Planning Authority shall be notified immediately and no further work shall be carried out until a method statement detailing a scheme for dealing with the suspect contamination has been submitted to and agreed in writing with the Local Planning Authority.

Reason: To ensure all contamination within the site is dealt with in accordance with Policy 30 of the South East Lincolnshire Local Plan 2011-2036.

In determining this application the authority has taken account of the guidance in paragraph 38 of the National Planning Policy Framework 2019 in order to seek to secure sustainable development that improves the economic, social and environmental conditions of the Borough.

Date: 21-Apr-2021

Mike Gildersleeves
Assistant Director – Planning
Boston Borough Council and East Lindsey District Council

Informatives

Black Sluice Internal Drainage Board comments

Rainfall Runoff

It is understood from the documentation provided that surface water from the development is to be discharged into a watercourse via an existing drainage system. As the applicant has stated that that the development will not increase the existing impermeable footprint, then the Board's consent is not required.

The applicant will be expected to ensure that any existing drainage system is of a suitable condition, including the status of any existing outfall headwall within the bank of the receiving watercourse (see Section 3).

Disposal of Foul or Dirty Water

It is understood that foul water from the development is to be discharged into an adjacent watercourse via a package or bio-treatment unit.

If this is the case, then the Boards prior written consent is required.

Discharge Outfalls

It is understood that the applicant is proposing a combined surface and treated water discharge outfall into the adjacent watercourse. If this is the case, then as per Section 1, the applicant will be expected to ensure that any existing outfall headwall structure is of a suitable condition.

If the structure requires replacement, then Section 23 Land Drainage consent will be required for any new or replacement structures to be placed in the bank.

Filling in or Culverting Watercourses

There are a number of watercourses, both open and piped, in the vicinity of the land concerned. Under Section 23 of the Land Drainage Act 1991, the prior written consent of the Board is required for any proposed permanent or temporary works or structures, within any watercourse, including infilling or diversion. This includes the replacement of any existing structures.

The applicant is reminded that under Common Law, the ownership and maintenance responsibility for any watercourse lies with the adjacent landowners, regardless of whether the watercourse is maintained by the Board.

Where a section of watercourse lies wholly within a land or property boundary, or lies alongside a highway, then the land or property owner is considered wholly responsible for the ownership and maintenance of that section of watercourse.

TOWN AND COUNTRY PLANNING ACT 1990

Appeals to the Secretary of State

- If you are aggrieved by the decision of your local planning authority to refuse permission for the proposed development or to grant it subject to conditions, then you can appeal to the Secretary of State under section 78 of the Town and Country Planning Act 1990.
- If this is a decision on a planning application relating to the same or substantially the same land and development as is already the subject of an enforcement notice, if you want to appeal against your local planning authority's decision on your application, then you must do so within 28 days of the date of this notice.
- If an enforcement notice is served relating to the same or substantially the same land and development as in your application and if you want to appeal against your local planning authority's decision on your application, then you must do so within 28 days of the date of service of the enforcement notice, or within 6 months [12 weeks in the case of a householder appeal] of the date of this notice, whichever period expires earlier.
- If this is a decision to refuse planning permission for a householder application, if you want to appeal against your local planning authority's decision then you must do so within 12 weeks of the date of this notice.
- If this is a decision to refuse planning permission for a minor commercial application, if you want to appeal against your local planning authority's decision then you must do so within 12 weeks of the date of this notice.
- If this is a decision to refuse express consent for the display of an advertisement, if you want to appeal against your local planning authority's decision then you must do so within 8 weeks of the date of receipt of this notice.
- If this is a decision to refuse consent for works to a tree protected by a Tree Preservation Order, if you want to appeal against your local planning authority's decision then you must do so within 28 days of the date of this notice.
- For all other decisions, if you want to appeal against your local planning authority's decision then you must do so within 6 months of the date of this notice.
- Appeals can be made online at: https://www.gov.uk/planning-inspectorate. If you are unable to access the online appeal form, please contact the Planning Inspectorate to obtain a paper copy of the appeal form on tel: 0303 444 5000.
- The Secretary of State can allow a longer period for giving notice of an appeal but will not normally be prepared to use this power unless there are special circumstances which excuse the delay in giving notice of appeal.
- The Secretary of State need not consider an appeal if it seems to the Secretary of State that the local planning authority could not have granted planning permission for the proposed development or could not have granted it without the conditions they imposed, having regard to the statutory requirements, to the provisions of any development order and to any directions given under a development order.
- If you intend to submit an appeal that you would like examined by inquiry then you must notify the Local Planning Authority and Planning Inspectorate (inquiryappeals@planninginspectorate.gov.uk) at least 10 days before submitting the appeal. Further details are on GOV.UK.
- Proposed Demolition The provision of Section 80 of the Building Act 1984 require that anyone who intends to demolish a building or structure shall give notice to the Council's Building Control department (01205 314295) beforehand, and work shall not commence until the Local Authority has given Notice under Section 81, or the relevant period of 6 weeks has expired.

THE IS A LEGAL DOCUMENT - PLEASE RETAIN IN A SAFE PLACE, PREFERABLY WITH YOUR DEEDS